Author’s response to reviews

Title: Case reports: three Novel variants in PCCA and PCCB Genes in Chinese Patients with propionic acidemia

Authors:

Qi Yang (15625056073@163.com)
Hong Xu (xunhongqinhai@126.com)
Jingsi Luo (490880259@qq.com)
Mengting Li (1024628723@qq.com)
Sheng Yi (342908980@qq.com)
Qinle Zhang (zhangqinlesklmg@163.com)
Guoxing Geng (345347209@qq.com)
Shihan Feng (fengshihansk@163.com)
Xin Fan (yangqisklmg126@126.com)

Version: 5 Date: 19 Mar 2020

Author’s response to reviews:

Dear Editor:

We greatly appreciate this opportunity to revise our manuscript. We would like to thank the reviewers for their kind comments regarding our work and for their efforts in reviewing this paper. We have carefully read the editor’s comments regarding our submission (MGTC-D-19-00561R4), and have responded point by point to each of the editor’s comments as listed below, and the revised portions in the paper. We feel that the editor’s comments have greatly helped us to strengthen our manuscript and we hope that the revised version will be acceptable for publication in BMC Medical Genetics.

Sincerely yours,

QI Yang.

Department of Genetic Metabolism
The Maternal and Child Health Hospital of Guangxi Zhuang Autonomous Region
Many thanks for including a link to [https://pan.baidu.com/s/1vDtsnFRe8E50AFlvvGmw] in the "Availability of Data and Materials" section, however please may you provide this in an international (or at least English language) repository?

Response: Thanks for the excellent suggestion. The datasets generated and analysed during the current study are available in the Mendeley repository, https://data.mendeley.com/datasets/dk3pfb8f5/3 (DOI: 10.17632/dk3pfb8f5.3).